2009-07-23 Pbar Department Meeting Thursday, July 23, 2009
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9:59 AM

his week

Transformers - done
Power Supplies
= Waiting for water to be turned back on at F23 (off for a week)
Shunts
IB7
Magnet is in
Final align was done yesterday
Collamator needs to be done.
DS end of beam pipe on IB7 was too short, so have to modify pipe that is
downstream of that. Starting work on that today
= Before pump down, need pipe done and collimator move complete.

Collimator
D:ISEP

= Being welded right now.
4-8GHz Tanks
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e A50/60 Bake

o By tomorrow at 8am will be ramped down.

Target Station

e SEM
e Target
e Lens

e Water Cage
Maintenance/repair
e RF
o Pat has been working on DRF
e PLC/Controls
o  Working on temp on thermistor chasiss for Debuncher filters.
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e Vacuum system
o Changes in two weeks
Surveys
e Inductance - done
e Activation - done
Water
o Leaks
o Turn off LCW pumps next week????
o Brazing done
o Leaks given to Mech Supt.
o IB7 - don't connect the water until it is flushed.
e Back Flush
e Check all water flowing
Antihydrogen part II
= Dipoles will be connected.
FESS/Infrastructure
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Dumb Waiters

Lights
Pre-Target Door
o Itisinstalled
o There is an air sampling hose on the wrong side, so it will need to be
extended.
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e Janitorial
Power Outage
e Sijte Wide Mon Jul 27

]
Admin Assistant
e Deb gone for 6 weeks
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